Author's response to reviews

Title: Use of quantification of vision by visual evoked potentials in visual disability assessment

Authors:

Jihoon Jeon Dr. (jjh786@hanmail.net)
Sungeun E Kyung Dr. (Kyungeye@gmail.com)

Version: 4 Date: 23 February 2012

Author's response to reviews:

I made changes in my manuscript.

Institutional review board exemption was obtained from the university medical Center, and the design of the study followed the principles of the Declaration of Helsinki.